Author's response to reviews

Title: Do gastrointestinal complaints increase the risk for subsequent medically certified long-term sickness absence? The HUSK study.

Authors:

Simon Øverland (Simon.Overland@iu.h.uib.no)
Marit Knapstad (Marit.Knapstad@hemil.uib.no)
Ingvar Wilhelmsen (Ingvar.Wilhelmsen@med.uib.no)
Arnstein Mykletun (Arnstein.Mykletun@uib.no)
Nick Glozier (Nick.Glozier@sydney.edu.au)

Version: 6 Date: 27 July 2011

Author's response to reviews: see over
Dear editors,

Thank you for checking through our manuscript. We have now attended to the formatting changes asked for. In detail, we have:

- Provided the complete affiliation information for all authors on the cover page.
- Provided names for all authors in ref #1, instead of the previous use of "et.al" in the reference list.
- Added email address for the corresponding author.
- Replaced roman numerals in tables with numbers.
- Replaced the green lines in the tables with black lines.
- Made some minor editorial/typographical corrections in various places in the manuscript.

All changes are marked using “track changes” in the document we now submit.

Sincerely, on behalf of the authors,

Simon Øverland
Corresponding author